Author’s response to reviews

Title: Self-Reported Psychopathy in the Middle East: A Cross-National Comparison across Egypt, Saudi Arabia, and the United States

Authors:
Robert Latzman (rlatzman@gsu.edu)
Ahmed Megreya (amegreya@qu.edu.qa)
Lisa Hecht (lhecht1@student.gsu.edu)
Joshua Miller (jdmiller@uga.edu)
Anne Winiarski (anne.winiarski@emory.edu)
Scott Lilienfeld (slilien@emory.edu)

Version: 1 Date: 18 Jul 2015

Author’s response to reviews:
As requested, appropriate additional information concerning human subjects protections as well as informed consent procedures have been added to the Methods section.

Further, the title page now includes the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.